Author response to reviewer comments

Thank you reviewers for taking the time to provide valuable feedback on the manuscript entitled “Cardiac Risk Stratification of the Liver Transplant Candidate: A Comprehensive Review” that is under consideration for publication in World Journal of Transplantation. We appreciate the thoughtful comments and suggestions provided to us as it has helped us improve our work significantly. We are grateful to participate in this author response process and have attempted to address each of the reviewers’ and editor’s comments to the best of our ability. Please see attached our point-to-point responses. My co-authors and I look forward to the editorial and reviewer reassessment.

Reviewer#1:
Specific Comments To Authors: Excellent review that covered all aspects related to preoperative assessment and postoperative outcome of liver transplant candidate from cardiac point of view

Author response
Thanks or your comments.

Reviewer#2:
Specific Comments To Authors: Interesting article addressing the important aspect of liver transplantation in people with end-stage liver disease (ESLD) who often suffer from multiple comorbidities, of which cardiovascular disease (CVD) is a major issue with prognostic implications in the peri-and post-transplant period.

Author response
Thanks or your comments.